Author’s response to reviews

Title: Correlates of Opium Use: A Study in Tribal Community of Arunachal Pradesh, India

Authors:

Himanshu K. Chaturvedi (chaturvedi_icmr@yahoo.com)
Jagadish Mahanta (jmahanta@hotmail.com)
Ram C. Bajpai (rambaipai@hotmail.com)
Arvind Pandey (arvindpandey@icmr.org.in)

Version: 3 Date: 28 February 2013

Author’s response to reviews:

Dear Sir/Madam,

This has reference to your e-mail dated 15 February, 2013 regarding the submission of manuscript (MS: 9564754986397376) title “Correlates of Opium Use: A Study in Tribal Community of Arunachal Pradesh, India” by Chaturvedi et al. As advised by the Editor, the English language of the manuscript has been edited. The revised manuscript is also prepared as per the format of the Journal. The revised manuscript is uploaded for your kind consideration.

With regards,

H. K. Chaturvedi
Scientist ‘E’
National Institute of Medical Statistics,
ICMR, Ansari Nagar, New Delhi – 110029